

Article



Structure Design and Heat Transfer Performance Analysis of a Novel Composite Phase Change Active Cooling Channel Wall for Hypersonic Aircraft

Weichen Li¹, Jieliang Zhao^{1,*}, Xiangbing Wu¹, Lulu Liang¹, Wenzhong Wang¹ and Shaoze Yan²

- ¹ School of Mechanical Engineering, Beijing Institute of Technology, Beijing 100081, China; 3120210421@bit.edu.cn (W.L.); 18801165131@163.com (X.W.); lianglulu12345@163.com (L.L.); wangwzhong@bit.edu.cn (W.W.)
- ² Division of Intelligent and Biomechanical Systems, State Key Laboratory of Tribology in Advanced Equipment, Department of Mechanical Engineering, Tsinghua University, Beijing 100084, China; yansz@mail.tsinghua.edu.cn
- * Correspondence: jielzhao@bit.edu.cn

Abstract: Efficient and stable heat dissipation structure is crucial for improving the convective heat transfer performance of thermal protection systems (TPSs) for hypersonic aircraft. However, the heat dissipation wall of the current TPS is limited by a single material and structure, inefficiently dissipating the large amount of accumulated heat generated during the high-speed maneuvering flight of hypersonic aircraft. Here, a convection cooling channel structure of TPS is proposed, which is an innovative multi-level structure inspired by the natural honeycomb. An active cooling channel (PCM-HC) is designed by using a variable-density topology optimization method and filled with phase change material (PCM). Numerical simulations are used to investigate the thermal performance of the PCM-HC wall, focusing on the influence of PCM properties, structural geometric parameters, and PCM types on heat transfer characteristics. The results demonstrate that the honeycomb-like convection cooling channel wall, combined with PCM latent heat of phase change, exhibits superior heat dissipation capability. With a heat flux input of 50 kW/ m^2 , the maximum temperature on the inner wall of PCM-HC is reduced by 12 K to 20 K. Different PCMs have opposing effects on heat transfer performance due to their distinct thermophysical properties. This work can provide a theoretical basis for the design of high-efficiency cooling channel, improving the heat dissipation performance in the TPS of hypersonic aircraft.

Keywords: active cooling thermal protection; honeycomb structure; topological optimization; phase change material; heat transfer performance

1. Introduction

Hypersonic aircraft have gradually become a research hotspot in the field of aerospace due to their high maneuverability [1] and strong penetration capability [2]. However, when the aircraft cruises at speeds exceeding 5 Ma in near space, it will be subjected to severe aerodynamic heating [3] and air viscosity friction [4], resulting in surface temperatures of over one thousand degrees Celsius [5,6]. Although most of the heat can be isolated by the Thermal Protection System (TPS) of the aircraft, the remaining aerodynamic heat transmitted to the inside of the structure will cause internal thermal strain and result in an increase in the temperature of the cabin, which will affect the normal use of the thermally sensitive equipment and change the mechanical properties of the structure's materials, thus affecting its strength and stiffness [7,8]. Therefore, it is necessary to develop a convective cooling structure that can both bear external load and actively cool internally to better dissipate the heat inside the aircraft's structure.

The fundamental structure of thermal protection systems comprises an outer high-temperature structural layer and an internal non-load-bearing thermal insulation layer [9,10].



Citation: Li, W.; Zhao, J.; Wu, X.; Liang, L.; Wang, W.; Yan, S. Structure Design and Heat Transfer Performance Analysis of a Novel Composite Phase Change Active Cooling Channel Wall for Hypersonic Aircraft. *Micromachines* **2024**, *15*, 623. https://doi.org/10.3390/mi15050623

Academic Editor: Stéphane Colin

Received: 31 March 2024 Revised: 30 April 2024 Accepted: 3 May 2024 Published: 6 May 2024



Copyright: © 2024 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). The structural layer, composed of high-temperature-resistant metals or composite materials, is designed to withstand external aerodynamic loads, while the thermal insulation layer, consisting of insulating materials applied as coatings and skins, serves to isolate the substantial heat generated by aerodynamic heating. Given that hypersonic vehicles endure prolonged exposure to high-heat-flux aerodynamic thermal environments, coupled with stringent requirements for flight speed and structural integrity, thermal protection structures often exhibit considerable thickness, thereby unavoidably introducing issues of increased mass that adversely impact flight velocity and escalate fuel consumption [11]. Existing convective cooling channels have been employed in scenarios characterized by high heat flux density, such as in the supersonic combustion ramjet engines of hypersonic vehicles [12], where adjustments to channel geometry and dimensions enhance heat dissipation capabilities. Similarly, hypersonic vehicles' surfaces confront severe thermal loading environments, prompting the integration of convective cooling channels into TPS to replace bulky thermal insulation materials with high-efficiency heat transfer mechanisms. This approach significantly aids in reducing vehicle mass while preserving structural load-bearing capabilities.

Presently, active TPS are predominantly employed in hypersonic vehicles, leveraging active cooling techniques to remove greater amounts of heat, thereby minimizing the limitations imposed by insufficient material heat resistance and mitigating structural damage to the vehicle [13]. According to different cooling methods, active TPS can be divided into three types: sweat cooling, thin film cooling, and convective cooling [14]. Liu et al. [15] designed a sweating-type TPS, which played a role in reducing the structural temperature to some extent, and the sweating cooling reduced the internal structure temperature of the test model by 50 K. Zhou et al. [16] experimentally verified the influence of low compressibility and the ratio of coolant to heat flux density on the thin film cooling performance of the model. Lee et al. [17] numerically studied the thermal performance of a pin fin heat sink and reduced its thermal resistance by 11% and weight by 30% by optimizing the fin density. Among these active cooling technologies, convective cooling, as a clean and renewable cooling method, has received particular attention and become one of the effective thermal protection methods for hypersonic aircraft in long-term service. In addition to the above features, one of the most attractive features of convective cooling (regenerative cooling) is the use of fuel as a coolant, which not only avoids coolant waste but also achieves efficient and environmentally friendly utilization of residual heat energy [18,19].

However, traditional convective cooling techniques also have certain problems, including low heat dissipation, large coolant consumption, and susceptibility to failure [20,21]. Eckert et al. [22] studied the coolant consumption of three cooling methods, namely impinging cooling, convective cooling, and thin film cooling, under the same mainstream conditions. It was found that convective cooling has the highest coolant consumption among the three cooling methods: under laminar flow conditions, convective cooling is 3 times higher than sweating cooling, and under turbulent flow conditions, it is 1.5 times higher than sweating cooling. This also results in a larger weight of the convective cooling system, and its application is mainly concentrated on heat sinks for miniature devices such as electronic components [23], posing significant technical challenges for TPS applications in spacecraft [24]. Thus far, researchers have conducted some studies on the application of convective cooling technology in the aerospace field, including the use of cold liquid hydrogen circuits for engine cooling in aerospace vehicles and its application in TPS for SSTO hot zones [25]. These strategies can enhance fluid perturbation and heat transfer in convective cooling structures, but they cannot fundamentally solve the aforementioned issues associated with convective cooling technology.

In order to enhance the heat dissipation capability of convective cooling channels and improve the load-bearing performance of the channels themselves, natural honeycomb structures have come into our focus. The honeycomb structure itself exhibits excellent mechanical properties, and moreover, the temperature of the honeycomb can maintain dynamic stability, varying between 25 and 45 °C according to environmental temperatures [26]. However, existing research has mostly focused on the macroscopic structure

3 of 26

characteristics and component properties of honeycombs. Zhang et al. characterized the multilevel structure of natural honeycombs at different scales in detail [27–29], and used ESEM, AFM, and other methods to observe and analyze the microstructure and composition of the honeycomb, revealing the micro-morphology of the layered composite material composed of beeswax and bee silk. Zhang also studied the principles of anisotropy and temperature control in honeycomb structures through experiments and numerical simulations, proposing a novel porous material design with a surface composite coating. By mimicking the microstructure of honeycombs, this design breaks through the limitation of only biomimicry from a macroscopic perspective. Duan [30,31] proposed honeycomb equivalent mechanical models and expansion performance models, analyzing the mechanical properties of honeycombs at different temperatures. It was found that bees improve the mechanical performance of honeycombs by cyclically using bee silk, which affects the internal heat transfer capacity of the honeycomb and thereby regulates the temperature of the honeycomb.

The honeycomb structure provides an excellent biomimetic object for the design of convective cooling channels. Recently, many scholars have conducted research on the biomimetic heat exchange structure of honeycombs. Lu [32] used a corrugated wall model to evaluate the heat transfer of microporous aluminum honeycomb under convective cooling in a compact heat exchanger, obtaining the optimal honeycomb cell morphology to achieve maximum overall heat transfer efficiency. Wen [33] studied the performance of sandwich metal honeycomb structures under forced air convection conditions through experiments and numerical methods, testing honeycomb structures made of different materials and with different topological structures. The results showed that the total pressure drop was related to the specific surface area density and the honeycomb cell shape, while the total heat transfer rate was related to the specific surface area density, cell shape, and thermal conductivity of the solid material. Pei et al. [34] proposed a honeycomb-shaped ultra-compact plate heat exchanger and studied its flow and heat transfer characteristics, obtaining the optimal geometric structure of the heat exchanger. Therefore, the honeycomb structure has great potential for application in the field of heat exchange, and it is necessary to break through the limitations of traditional porous materials and design convective cooling structures from the perspective of honeycomb microstructure in order to achieve better heat transfer effects.

With the help of microstructure features, the honeycomb achieves a more reasonable heat distribution through heat transfer, thereby improving the heat exchange capacity between the honeycomb walls and the external environment. In order to better apply the honeycomb's ability to convective cooling channel wall heat transfer, the phase change latent heat characteristics of phase change materials (PCM) can effectively delay the heat transfer on the wall surface [35], allowing more heat to accumulate on the aerodynamic heated flow side of the channel and avoiding transfer to the far end, thus improving the heat transfer efficiency with the fuel medium. PCM has been widely used in fields such as aerospace and battery thermal management. Kim et al. [36] proposed a new type of spacecraft thermal control hardware combined with PCM and studied the thermal carrying capacity of PCM through simulation. Wang et al. [37] studied the influence of composite PCMs composed of expanded graphite and paraffin with different melting points on battery thermal management performance at different phase change temperatures, and improved the efficiency of PCM through numerical optimization.

In recent years, topology optimization methods have gradually become a new way of designing heat transfer structures [38–40]. By satisfying conditions such as objective functions and design variables, topology optimization seeks the optimal material layout in the design domain to achieve a certain performance optimum [41]. Tang et al. [38] proposed a topology optimization model for steady-state nonlinear heat conduction structures under large temperature gradients and found that solid material variability affects topology optimization configurations. Zhuang et al. [42] studied the topology optimization problem of transient heat conduction structures with the integration of temperature gradients at fixed time intervals as the objective function. Xia et al. [43] combined the level set method with bi-directional evolutionary structural optimization to perform heat transfer topology optimization on a flat plate. However, the aforementioned studies commonly focus on optimizing thermal parameters to enhance the thermal characteristics of structures, with little consideration given to the load-bearing properties of thermal structures. In the field of statistics, there is a paucity of research related to topology optimization with the objective of minimizing flexibility.

Topology optimization has also found application in the design of biomimetic honeycomb heat transfer structures. Zhou [44] developed a honeycomb structure with cooling channels, employing topology optimization based on parameters such as thermal shielding and dissipation to study its thermomechanical behavior. Li et al. [45] proposed a heterogeneous honeycomb structure and established a single-ring topology optimization model, successfully reducing its thermal elastic deformation. Huang et al. [46], through the MTO optimization method, adjusted the thermal conductivity and convective area to design a honeycomb structure with high cooling performance. While honeycomb structures hold great potential in the field of heat transfer structures, existing topology optimization designs only mimic the macrostructure of natural honeycombs, namely by using homogeneous pore walls without microstructure to imitate their hexagonal structure. It is necessary to break through the limitations of traditional porous materials and conduct topology optimization design from the perspective of honeycomb microstructure to achieve better heat transfer efficiency.

In order to better utilize the biomimetic value of honeycomb microstructure, this study combines biomimetics with topology optimization methods to biomimetically design honeycomb structures at multiple scales, considering both the macroscopic and microscopic features of natural honeycombs. By integrating PCM with the honeycomb structure and utilizing the latent heat storage characteristics of PCM, a novel composite phase change material-honeycomb channel (PCM-HC) is designed. The thermal performance of the PCM-HC's wall is investigated, taking into account its heat transfer properties.

The remaining sections of this paper are organized as follows. Section 2 presents the construction of a biomimetic honeycomb convective cooling channel (HC) and outlines the micro-design principles for HC. Section 3 introduces a density-based topology optimization method based on the SIMP interpolation model, which is used for two-dimensional cross-sectional topology optimization of HC. Based on this, a multi-scale structural design method called PCM-HC is proposed. In Section 4, a transient heat transfer analysis model for HC and PCM-HC is established using the finite element method (FE). Section 5 analyzes the influence of PCM presence, geometric parameter variations, and the use of different PCMs on the wall heat transfer performance of PCM-HC. Finally, Section 6 provides a summary of the main findings of this study.

2. Biomimetic Honeycomb HC Model

Honeycombs possess distinctive structural characteristics that are highly suitable for biomimetic emulation. For instance, they are designed with minimal material usage while maintaining stability, making them an ideal choice for porous structures. Additionally, the inherent insulation capability of honeycombs is worthy of extensive attention. Honeybees employ an ingenious architectural strategy by incorporating propolis as an additive into beeswax, forming the cell walls of honeycombs. This facilitates the regulation of hive temperature in response to a decrease in external environmental temperature. Furthermore, bees cover the surface of their nests with a detachable covering, which serves as passive insulation to regulate temperature. This mechanism helps prevent heat loss from the interior and protects against excessive external heat disturbance [47]. Therefore, at the macroscopic level, the hexagonal shape of the honeycomb is chosen as its overall configuration to enhance the mechanical strength of the convective cooling channel. Building upon this, at the microscopic level, the structural characteristics of biomimetic honeycomb cell walls, resembling layered composite materials, are utilized, with PCM filling the channel interstices, aiming to improve the heat transfer performance of the channel.

2.1. HC Model Construction

As shown in Figure 1c, leveraging the macroscopic hexagonal configuration of natural honeycombs, a design model of a honeycomb-inspired convective cooling channel, denoted as HC, was proposed based on the multi-level structure of honeycomb. The HC has a hexagonal shape with dimensions of $D \times H \times T$ (20 mm \times 10 mm \times 2 mm) and includes internal cooling channels for fuel flow to achieve convective heat transfer. In the application scenario of hypersonic variant aircraft nose cones, multiple HC units and other channel structures including stretching and bending deformation modules (Figure 1b) are connected to the internal skeleton structure of the aircraft morphing nose cone (Figure 1a) to remove the aerodynamic heat conducted from the skin to the internal structure. This design strategy can effectively improve the heat management performance of the hypersonic aircraft. Subsequent sections primarily focus on the heat transfer performance analysis of the biomimetic honeycomb convective cooling channel. The analysis excludes consideration of the remaining modified duct structures based on variant head cones.



Figure 1. HC model diagram. (a) The distribution pattern of HC on the surface of the morphing nose cone. (b) The connection method between HC and other functional structures. (c) The model structure of HC, with dimensions specified as $D \times H \times T$ (20 mm × 10 mm × 2 mm).

2.2. Micro-Design Principles of HC

The pore walls of natural honeycombs consist of two parts: the initial layer and the adhesive layer. The initial layer corresponds to the new honeycomb constructed during the initial stages of use, while the adhesive layer gradually forms during the honeycomb's usage. Consequently, the cell walls of natural honeycombs are considered as fiber composite materials composed of beeswax and bee silk. Taking into account the microstructure and mechanical properties of natural honeycombs, bees enhance the thermal stability of honeycombs by adding bee silk to the surface of new honeycomb walls [28]. Therefore, the design principles for the microstructure of HC walls are illustrated in Figure 2, mimicking the microstructural characteristics of natural honeycombs for HC wall design. Firstly, suitable metal materials are chosen to fabricate HC channels, representing the base material of natural honeycombs, beeswax. Subsequently, utilizing topology optimization methods, parts of the base material are reasonably removed and replaced with other materials (i.e., PCM) to form a micro-layered composite structure similar to honeycomb pore walls.



6 of 26



Figure 2. The micro-design method for HC. The HC originates from the new honeycomb, whereas the old honeycomb consists of an initial layer and adhesive layer. Utilizing metal materials as the substrate, and incorporating additional materials, a layered composite structure is designed for PCM-HC, mimicking the layered composite structure of the old honeycomb.

3. Topology Optimization Design of PCM-HC

3.1. Topology Optimization Method

A density-based method for topology optimization of continuum structures was employed. The optimization objective was to maximize stiffness while maintaining the optimal HC bearing characteristics by removing some wall material and filling it with PCM to improve the heat transfer capacity of the wall. The density-based method assumes that the material density of each element is different. Material optimization is achieved through independent variation of the relative density of the material. The mathematical model is given as follows [48]:

Find :
$$\rho = \{\rho_1, \rho_2, \cdots, \rho_n\} \in R$$
,
Minimize : $C = \frac{1}{2}U^T K U$,
Subject to : $K U = F$, (1)
 $\rho_i \in [0, 1], \forall_i$,
 $g = \sum_i \rho_i - V_0 \le 0$,

where ρ , R, U, K, F, and V_0 are the design variable, the design domain, the displacement matrix, the stiffness matrix, the load matrix, and the target optimized volume, respectively.

In topology optimization, the objective of maximizing stiffness is translated into minimizing the elastic strain energy expressed by the objective function *C* using matrix notation.

The material interpolation models can be classified into two methods: the RAMP method and the SIMP method. These two models are quite similar, but the SIMP method

exerts slightly stronger penalization on the element density. The material interpolation model based on the SIMP formulation is given as follows [49]:

$$E_{i} = p(\eta_{i})E_{i_{0}} = \eta_{i}^{p}E_{i_{0}}$$
⁽²⁾

where E_{i_0} is the elastic modulus of the *i*-th element in the fully dense state ($\eta_i = 1$), $P(\eta_i)$ denotes the interpolation function, and *p* is the penalty coefficient, which is set to 3.0 [48].

Under the condition of inertial load being overloaded, the classical SIMP model may lead to distortion in low-density elements due to the mismatch between material mass and stiffness models. To address this issue, a polynomial interpolation model can be employed to mitigate the occurrence of element distortion:

$$E_i = [(1 - \omega)\eta_i^p + \omega\eta_i]E_{i_0} \tag{3}$$

where ω is the weight coefficient of the linear term. Therefore, the ratio between the penalty coefficient of element mass and stiffness is defined as follows:

$$r_{mK} = \frac{1}{(1-\omega)\eta_i^{p-1} + \omega} \tag{4}$$

By incorporating ω in the denominator, regardless of the value of η_i within the range of [0, 1], r_{mK} remains finite in size, resolving the issue of element distortion in the SIMP model.

3.2. Topology Optimization Process

The HC channel is a symmetrical structure (Figure 3a) that can be split in half along the symmetry plane, while the 2D cross section gradually lengthens into an arch-like structure along the axial direction. The stress distribution of the 2D cross section is similar, allowing the 3D problem to be simplified to a 2D problem. Therefore, the optimization of the structural topology was carried out on two cross sections of the HC channel, namely at the end and middle sections, and the result was extrapolated to the 3D topology structure (Figure 3b).

Utilizing the topology optimization module (ATOM) in Abaqus for structural topology optimization, a 2D cross-sectional topology optimization model was established. The bottom surfaces of both end sections and the midsection were fully constrained, while the upper surface was subjected to a uniformly distributed load F. Considering the practical application scenario of PCM-HC heat pipes, the model distinguished between the design domain and non-design domain. A boundary layer within 0.2 mm of the outer contour of the wall was frozen (Figure 3c,d). Within ATOM, stress, volume fraction, and weighted strain energy were initially defined as responses to be constrained or targeted. Stress constraints were based on the compressive strength of SUS304 (515 MPa), with an upper limit of 500 MPa. Volume fraction constraints were set as design variables, with volume fraction gradients of 40%, 55%, and 70% (representing the ratio of optimized volume to original volume). The weighted strain energy was selected as the objective function, with the optimization direction set to minimize. Simultaneously, in the geometric constraint module, a minimum size constraint of 0.1 was defined to prevent the generation of overly thin structures, thereby achieving more uniformly distributed material distribution in the optimized results.

The static stress contour maps of the end and middle sections of the HC channel in Figure 3e,f show that the stress concentration is mainly located around the middle region of the upper wall surface and at the diagonal intersection between the upper wall and side walls. These regions should be preserved as the main load-bearing parts in the design of the PCM-HC channel. As the bottom surface of the end and middle sections serves as the farthest point of thermal conduction and exhibits low stress, this area will become the main removal region in the topology optimization process and is therefore designated as the

non-design domain. Additionally, the maximum stress on the compressed side wall of the middle section is approximately 5.2 times greater than that of the end section. Therefore, the topology optimization result of the middle section was selected as the overall design reference for the PCM-HC channel.



Figure 3. The topology optimization analysis process of HC is transformed from a 3D problem to a 2D problem. (**a**,**b**) HC model simplification process. The HC structure is symmetrical and includes an internal channel for coolant flow, which is simplified into a 2D arch-shaped problem. (**c**,**d**) Topological optimization model. Two cross-sections, one located at the end and another at the middle of the structure, are selected for topology optimization. (**e**,**f**) Stress contour plots for the end section and middle section are presented. The difference of color in different regions of the cross section represents the difference of stress magnitude. The maximum stress values for the end section and middle section are 0.116 MPa and 0.602 MPa, respectively.

In Figure 4a–d, topology optimization is performed with volume fraction gradients set as 100%, 70%, 55%, and 40%, indicating that the optimization mainly focuses on areas of low stress in the structural design region. Therefore, topology optimization primarily optimizes the lowest-performing units in the overall structural performance area by removing inefficient units in order to improve the overall performance and mechanical properties of the structure. As the volume fraction changes, the outline of the main loadbearing structure becomes clearer: for the upper wall surface, the material reduction is mainly concentrated in the middle Section 1, with a trapezoidal shape, and symmetrically distributed along the cross-sectional axis. For the side wall surface, with a volume fraction of 40%, a "Z"-shaped support structure is clearly observed in area 2, indicating that this design method can enhance the lateral load-bearing capacity of the overall structure.



Figure 4. Topology optimization results of the middle section. (**a**) The initial structure diagram of the middle section. (**b**–**d**) Topology optimization results under volume fraction constraints of 70%, 55%, and 40%. Under a volume fraction constraint of 40%, region 1 exhibits a trapezoidal structure, while region 2 forms a "Z"-shaped pattern. (**e**) Variation of strain energy with iterations at different volume fractions.

In order to compare the mechanical strength of HC sections at different volume fractions, Figure 4e illustrates the variation of strain energy with topology optimization iterations for volume fractions of 70%, 55%, and 40%. Strain energy serves as a measure of structural stiffness, where minimizing strain energy is equivalent to maximizing overall stiffness. As observed in the figure, with an increase in iterations, the optimization effectiveness of HC sections' stiffness improves progressively. At the conclusion of optimization, the strain energy of HC sections with volume fractions of 40% and 55% exceeds that of HC sections with a volume fraction of 70% by 56.2% and 20.4%, respectively. Thus, as the volume fraction decreases, the structural stiffness of HC sections diminishes. Consequently, when the volume fraction is 70%, the mechanical strength is optimal.

3.3. Determination of PCM-HC Model

Based on the topology optimization analysis results of the middle section, a gradual removal of the substrate material SUS304 (Figure 5b) along the cross-sectional direction of HC (Figure 5a) is performed. Subsequently, the cavity is filled with PCM to obtain the PCM-HC model (Figure 5c). In Figure 5d, the belly plate thickness (*N*) on the side wall varies between 0.15 mm, 0.30 mm, 0.45 mm, and 0.60 mm, while the PCM region on the upper wall has a trapezoidal shape. The characteristic parameters of the trapezoidal PCM region are determined based on the trapezoid area formula S = (a + b)h/2. The trapezoidal PCM cross section characteristic vector is defined as follows:

$$\mathbf{W}_{i} = (\alpha_{i}, \beta_{i}, h_{i}) = (\frac{a_{i}}{h_{i}}, \frac{b_{i}}{h_{i}}, h_{i}), \quad i = 1, 2, \dots, n$$
(5)



where a_i is the length of the upper base of the trapezoid, b_i is the length of the lower base of the trapezoid, and h_i is the height of the trapezoid.

Figure 5. The process of establishing the PCM-HC model. The HC structure (**a**) is modified by removing a portion of SUS304 material (**b**) and filling it with PCM to obtain a PCM-HC structure (**c**). (**d**) The trapezoidal PCM area height is *h*, the upper bottom length is *a*, the lower bottom length is *b*, and the web thickness is *N*.

At this point, the area of the trapezoidal PCM region can be expressed as follows:

$$S = \frac{1}{2}(\alpha_i + \beta_i)h_i^2 \tag{6}$$

Keeping the area of the trapezoidal PCM region $S = 6 \text{ mm}^2$ constant, let $W_1 = (\frac{10}{3}, 5, \frac{6}{5})$, $W_2 = (\frac{5}{3}, \frac{20}{3}, \frac{6}{5})$, $W_3 = (4, 8, 1)$, $W_4 = (2, 10, 1)$.

When hypersonic vehicles cruise in near-space, they will experience severe aerodynamic loads, imposing significant structural demands on the internal convective cooling channels. The HC and PCM-HC model, inspired by the high mechanical performance of honeycomb structures, are compared with traditional square channels through a simulation of their mechanical properties to verify the structural load-bearing capacity. The square channels maintained the same load-bearing area and internal volume as the HC and PCM-HC models (Figure 6a).



Figure 6. The finite element models and mechanical property analysis of the channel. (**a**) The finite element models of the HC channel, PCM-HC channel, and square channel maintain the same load-bearing area and cavity volume. (**b**) The variation of load capacity with displacement for the HC channel, PCM-HC channel, and square channel under axial compression between rigid plates is investigated.

Mechanical analysis was conducted using the commercial finite element software Abaqus 2021. As depicted in Figure 6a, the finite element model of the channel was placed between upper and lower rigid plates. The lower end of the rigid plate was fixed in all degrees of freedom, while the upper rigid plate compressed downward at a constant speed of 0.5 mm/s. The contact between the upper and lower rigid plates and the channel model was defined as general contact, with tangential penalty contact, a friction coefficient of 0.2, normal hard contact, and allowed separation after contact. The material of channels was defined as SUS304, with material parameters listed in Table 1. HC and square channels are discretized using eight-node linear hexahedral elements (C3D8R), while the PCM-HC channel employs ten-node quadratic tetrahedral elements (C3D10) for discretization, culminating in the completion of the computational solution.

Table 1. Material parameters of SUS304.

Material	$ ho/{ m g}\cdot{ m cm}^{-3}$	E/Gpa	σ /Mpa	v
SUS304	7.93	204	800	0.285

The force-displacement curves under axial compression for the channel models depicted in Figure 6b reveal three distinct stages of deformation, namely elastic, yielding, and plastic deformation, for HC, PCM-HC, and square channels. With increasing axial displacement, the mechanical characteristic curves of the models exhibit a gradual upward trend. However, HC demonstrates significantly higher axial load-bearing capacity compared to square and PCM-HC channels, with peak test forces at yield strengths of 34.5 kN, 20.5 kN, and 19.0 kN, respectively. HC exhibits a 68% increase in load-bearing capacity over the square channel under identical bearing area and heat transfer area conditions. Even after removing some material to accommodate PCM, HC maintains comparable load-bearing performance to the traditional square channel.

4. Research on PCM-HC Performance

4.1. Physical Model Description

Due to the uniform heat flux conditions imposed on the upper surface of the convective cooling channel, both the HC and PCM-HC channels, as symmetric three-dimensional structures, can be simplified into a two-dimensional cross-sectional problem, where heat conduction along the length of the channel can offset each other. Under the same heat flux input conditions, if the temperature of the walls surrounding the channel is lower, the heat transfer burden on the cooling medium inside the channel is reduced. Establishing a two-dimensional cross-sectional model of the channel, the thermal distribution on the walls is studied in the absence of coolant flow, aiming to minimize the heat transfer burden on the convective cooling channel as much as possible.

To better simulate the practical application scenario of convective cooling channels, a conventional multi-layer TPS structure is introduced above the HC and PCM-HC structures. A thermal load is applied to the outer surface of the system to simulate the aerothermal load experienced by hypersonic aircraft during flight. The multi-layer TPS consists of an outer high-temperature structural layer and an inner non-load-bearing insulation layer, as shown in Figure 7. The upper layer, depicted in earthy yellow, is made of C/SiC composite high-temperature material to withstand the external thermal load Q_{in} , with a thickness of 2 mm. The middle layer, shown in dark blue, utilizes Saffil alumina fiber insulation material to isolate the significant heat generated by aerothermal effects, with a thickness of 10 mm. The lower layer, represented in deep green, serves as the convective cooling channel with a base material of SUS304. The gray area added in Figure 7b represents the filled PCM. The thermal and physical properties of C/SiC and Saffil refer to relevant literature [50,51], with some parameters varying with temperature.

For the HC model, the heat transfer modes considered include thermal radiation Q_{rad} on the top surface and internal heat conduction Q_{cond} within the structure. For the PCM-HC model, the heat transfer modes include thermal radiation Q_{rad} on the top surface, internal heat conduction, and convective heat transfer Q_{con} which arises only within the PCM due to the phase change process. In addition, the contact thermal resistance between different material interfaces is ignored, and each contact surface is treated as an ideal heat transfer surface. Assuming the liquid phase of the PCM behaves as an incompressible fluid, the volumetric force effects resulting from PCM phase change are neglected [52,53].



Initial-temp=273.15K

Initial-temp=273.15K

Figure 7. Transient heat transfer analysis models for HC and PCM-HC. In HC (**a**) and PCM-HC (**b**), Q_{in} represents the input heat flux load, Q_{rad} denotes the surface thermal radiation of the system, Q_{cond} signifies the internal heat conduction within the system, and Q_{con} represents the convective heat transfer within the PCM. The arrows represent the direction of heat transfer. The initial temperature of the system is 273.15 K.

4.2. Heat Transfer Mathematical Model

The following equations represent the governing equations for HC and PCM-HC systems, including continuity, energy, and momentum equations, which are described as follows:

(a) Continuity equation [54]:

$$\frac{\partial \rho}{\partial t} + \nabla(\rho \vec{v}) = 0 \tag{7}$$

where ρ is density and \vec{v} is is velocity vector.

(b) Energy equation [55–57]:

$$\frac{\partial(\rho E)}{\partial t} + \nabla(\rho \vec{v} E) = \nabla(k \nabla T) + S$$
(8)

where *E* is the total enthalpy, *k* is the thermal conductivity coefficient, *T* is temperature, and *S* is the energy source term.

Total enthalpy *E* can also be represented as follows:

$$E = E_0 + \Delta E \tag{9}$$

where E_0 is sensible enthalpy and ΔE is latent enthalpy.

When the PCM is in a solid state, ΔE is 0. When the PCM is in a liquid state, ΔE is *L*, where *L* is the latent heat of the PCM. When the PCM is in a semi-solid or mushy state,

 ΔE takes on values between 0 and 1. The liquid fraction function δ , which is temperaturedependent, is defined as follows [58]:

$$\delta(T) = \frac{\Delta E}{L} = \begin{cases} 0, T < T_s \\ \frac{(T - T_s)}{(T_l - T_s)}, T_s \le T \le T_l \\ 1, T > T_l \end{cases}$$
(10)

where T_s denotes the lower limit of the phase transition temperature range and T_l denotes the upper limit of the phase transition temperature range.

In Equation (8), S is equal to the sum of the internal heat source and radiative flux, i.e.,

$$S = \nabla(\rho \vec{v} \Delta E) + \nabla Q_{rad} \tag{11}$$

When there is no internal heat source, the energy source term *S* is equal to the radiative flux. $\vec{Q_{rad}}$ represents the presence of outward thermal radiation on the upper surface of the system, which is defined by the following boundary condition equation [56]:

$$\vec{Q_{rad}} = k \frac{\partial T}{\partial y} = Q_{in} - \sigma \varepsilon (T_w^4 - T_e^4)$$
(12)

where σ is the Stefan–Boltzmann constant, $\sigma = 5.67 \times 10^{-8} \text{ W/(m^2 \cdot K^4)}$. Q_{in} is input heat flux density, ε is radiance, and T_w is wall temperature. T_e is ambient temperature, according to the International Standard Atmosphere (ISA), and values are determined for different flight altitudes.

(c) Momentum equation [54]:

$$\frac{\partial \vec{v}}{\partial t} + \nabla(\vec{v}\,\vec{v}) = -\frac{1}{p}\nabla P + \frac{1}{p}\nabla^2(\mu\nabla\vec{v}) + \vec{F}$$
(13)

where μ is dynamic viscosity, *P* is pressure, and *F* is momentum source term.

When PCM undergoes a liquid phase change, the expression of dynamic viscosity μ is as follows [59,60]:

$$\mu = 0.001 e^{(A + \frac{B}{T})} \tag{14}$$

where *A* is −4.25 and *B* is 1790 K.

Furthermore, in the analysis of transient heat transfer problems involving PCM, the enthalpy-porosity method is employed to calculate the phase change process [61]. In Equation (13), the momentum source term \vec{F} is defined as follows:

$$\vec{F} = \frac{c_m (1-\delta)^2}{(\delta^3 + \eta)} \vec{v},$$
(15)

where c_m is a mushy zone constant that is between 10^5 and 10^7 [62,63]. The introduction of η is to prevent division by zero. The function δ , mentioned earlier, represents the temperature-dependent liquid fraction.

4.3. Numerical Simulation Program

Transient heat transfer analysis of HC and PCM-HC was conducted using the Abaqus finite element software. Structured grids were employed, and a series of 2D models were established to analyze the heat transfer capability of the convective cooling channel when using PCM. The grid type used was DC2D4, with local refinement near the PCM region. The total number of grids was 5059. By solving the two-dimensional equilibrium equations at each node, the temperature distribution at various nodes was calculated, and a temperature contour plot of the model was generated.

As for the boundary conditions, a uniform heat flux load Q_{in} was applied to the top surface of the system, with a loading time of 600 s. Additional time calculations were performed to obtain the maximum temperature on the bottom wall of the channel. The outer surface is subjected to aerodynamic loads, while the two sides and bottom are situated within the aircraft with minimal gap areas where convection effects can be neglected. Therefore, both sides and the bottom of the system are set as adiabatic boundary conditions [56]. Furthermore, there was outward thermal radiation on the top surface with an emissivity (ϵ) value of 0.85. Additionally, the initial temperature condition for the entire system was set as 273.15 K.

5. Result and Discussion

Firstly, a comparison was made between the heat transfer characteristics of the convective cooling channel with and without the use of PCM filling. Then, in the presence of PCM, the combined effects of two parameters, namely the trapezoidal PCM cross section characteristic vector (**W**) and the belly plate thickness (*N*), were evaluated. In this study, five input heat flux densities were considered (30 kW/m^2 , 35 kW/m^2 , 40 kW/m^2 , 45 kW/m^2 , and 50 kW/m^2). Lastly, three types of PCM, namely Erythritol, Urea, and Parafin composite phase change material (CPCM), were selected to investigate their respective impacts on heat transfer.

5.1. Effect of PCM on Heat Transfer Performance

In this section, the heat transfer performance of convective cooling channels with and without PCM filling was compared for five different input heat flux densities (30 kW/m^2 , 35 kW/m^2 , 40 kW/m^2 , 45 kW/m^2 , and 50 kW/m^2). Two channel system models were established to compare the effect of PCM. The only difference between them was that the HC channel consisted entirely of SUS304 while the PCM-HC channel was composed of SUS304 and Erythritol. The input heat flux was uniformly applied to the surface of the system, and heat was transferred within the TPS and channel walls primarily through conduction and radiation, as detailed in Section 4.1. It should be noted that in this study, the trapezoidal PCM cross section characteristic vector was $(\frac{10}{3}, 5, \frac{6}{5})$, and the belly plate thickness was 0.30 mm.

Figure 8 illustrates the wall temperature distribution at different time intervals in the PCM-HC channel under a heat flux load of 50 kW/m^2 . In general, the heat flow is applied from the upper wall surface, resulting in a gradual decrease in wall temperature from the proximal end of the heat flow input to the distal end. Under the same heat transfer distance, the rate of heat transfer is dependent on the temperature difference between the object and its thermal conductivity, which can be explained by Fourier's law ($q = k\Delta T/L$, where q is the transferred heat flux, k is the material's thermal conductivity, ΔT is the temperature difference, and L is the distance between the heat exchange surfaces). At 510 s, the heat is uniformly diffused from the upper wall surface to both side wall surfaces, with a temperature difference of 14.2 K (Figure 8a). Subsequently, as the PCM reaches its phase change temperature, a significant amount of heat is absorbed during the phase change process. The heat transfer in the trapezoidal PCM region differs from the surrounding area, resulting in lower temperatures and a decreased rate of temperature rise, leading to almost constant temperatures (Figure 8b). At the same time, the PCM on the side walls also undergoes phase change and absorbs heat, resulting in a noticeable delay in temperature rise and a decrease in temperature gradient within the side walls (Figure 8c). At 540 s and 570 s, the temperature difference between the highest and lowest temperatures is reduced to 12.5 K and 13 K, respectively, as the phase change latent heat gradually decreases and the heat dissipation capacity reaches its limit. At 600 s, the PCM completes the phase change process, and the rate of temperature rise gradually recovers, with a reduced temperature difference of 8.8 K between the highest and lowest temperatures (Figure 8d).



Figure 8. The wall temperature distribution of PCM-HC varies with time under a heat flux load of 50 kW/m². (**a**–**d**) are 510 s, 540 s, 570 s, and 600 s, respectively. The arrows represent the order of time change. **W** = $(\frac{10}{3}, 5, \frac{6}{5}), N = 0.30$ mn.

As shown in Figure 9, the temperature variations along the AB path at 570 s and the CD path at 590 s under a heat flux load of 50 kW/m^2 are selected to demonstrate the thermal dissipation capability of PCM in the PCM-HC system. In the upper wall surface (Figure 9a), in the non-PCM region, the heat conduction rate is uniform and decreases slowly. However, in the PCM region, the temperature is significantly lower than in the non-PCM region, and the rate of temperature change accelerates. The temperature exhibits a decreasing trend followed by an increasing trend, with the lowest temperature dropping by nearly 5 K compared to the highest temperature in the non-PCM region. Additionally, compared to the HC channel without PCM, the PCM-HC system exhibits an average temperature reduction of 10 K to 15 K. Similarly, the side wall surface (Figure 9b) displays the same pattern of temperature variation. Due to the smaller area covered by PCM, the temperature drop in the PCM region is less pronounced. Nevertheless, it still demonstrates superior thermal dissipation capability compared to HC alone, with an average temperature reduction of 13 K. Thus, in the PCM-HC system, heat conduction near the PCM is slowed down, resulting in lower temperatures compared to the same location in the HC system and impeding heat transfer.

Figure 10a,b illustrate the temperature variations along the EF path (upper wall surface) and CD path (side wall surface), respectively, at the end of the heat flux loading for HC and PCM-HC under a 50 kW/m² heat flux input (at 600 s). These paths represent the inner wall surfaces involved in convective heat transfer with the fuel coolant, and a lower temperature distribution is desirable to reduce the burden of heat transfer with the fuel coolant. On the upper wall surface, both HC and PCM-HC exhibit a similar temperature trend, gradually increasing from the sides towards the center, with a temperature difference of approximately 4 K. This is because as the heat flux loading time increases, the phase change process of PCM is completed, leading to a loss in its ability to block heat transfer. The overall temperature of PCM-HC on the upper wall surface is about 12 K lower than that of HC. On the side wall surface, the temperature decrease in PCM-HC is faster than HC in the first half, while in the second half, the temperature decrease in PCM-HC is slower than HC. This is because PCM in the side wall surface is undergoing a phase change, absorbing a significant amount of heat due to latent heat, which attracts the heat flux from the upper wall surface and hinders heat conduction to the lower wall surface. The overall temperature of PCM-HC on the side wall surface is approximately 15 K lower than that of HC.



Figure 9. The temperature distribution along a specified path at a specific moment with and without PCM was examined. (**a**) The red line AB indicates the temperature distribution along the vertical trapezoidal PCM at 570 s on the upper wall surface. (**b**) The red line CD indicates the temperature distribution along the horizontal PCM at 590 s on the side wall surface.



Figure 10. The effect of PCM on the heat transfer performance of HC and PCM-HC. (**a**) The temperature distribution of the upper wall (EF red line). (**b**) The temperature distribution of the side wall (GH red line). (**c**) With the increase in heat flux input ($Q_{in} = 30, 35, 40, 45, and 50 \text{ kW/m}^2$), the maximum temperature of the lower wall changes. (**d**) $Q_{in} = 30, 40, and 50 \text{ kW/m}^2$; the temperature variation of the lower wall during the entire loading and unloading process after 600 s was recorded.

Figure 10c shows the variation of the maximum temperature on the lower wall surface with and without PCM in the convective cooling channel. As shown in the figure, as the heat flux input increases, the temperature difference between HC and PCM-HC undergoes a change from almost zero to gradually increasing and then maintaining a certain level. This is because when the heat flux input is small, within the 600 s loading time, PCM has not reached its phase change temperature and therefore cannot play a role in heat dissipation. With the increase in heat flux input, after the end of the 600 s heat flux loading, the remaining heat supply enables PCM to slowly reach its phase change temperature, thus exerting a certain heat dissipation effect. Finally, when the heat flux input continues to increase to a level that can quickly meet the latent heat of PCM's phase change, PCM fully exerts its heat dissipation effect. When the heat flux input is 45 kW/m², the heat dissipation effect is close to saturation, and the maximum temperature on the lower wall surface with PCM is about 20 K lower than that without PCM. This trend indicates that different scenarios of convective cooling channels correspond to different input heat fluxes and heating durations. Thus, it is necessary to select PCM with a suitable phase change temperature according to the specific requirements to fully utilize its heat dissipation capability.

Figure 10d illustrates the variation trend of the lower wall temperature during the entire loading and unloading process of PCM-HC and HC at different heat flux inputs for 600 s. As the heat flux input increases, both with and without the heat flow input, PCM exhibits good thermal conductivity. When the heat flux input is relatively small, there is no significant difference in temperature between the cases with and without PCM throughout the process, since the PCM has not reached its phase change temperature. When the heat flux input increases to 50 kW/m^2 , the highest temperature of the lower wall decreases from 402.7 K to 389.1 K with PCM under heat flow input, and from 430.8 K to 411.0 K without PCM under heat flow input. This reveals that the use of PCM to enhance the heat conduction of the wall in the convective cooling channel is suitable only for high heat flux inputs. Meanwhile, in the section without heat flow input, the temperature begins to drop after reaching its maximum value. As the temperature approaches the phase change temperature of the PCM, the PCM begins to store energy in the form of latent heat, and the temperature remains almost unchanged for a duration of 2000 s, delaying the temperature drop. After the PCM fully solidifies, further heat conduction causes the temperature to continue to decrease, resulting in a higher temperature of the channel with PCM than that without PCM, with a maximum temperature difference of 7.5 K. This implies that the solidification phase change of the PCM during temperature drop is crucial for maintaining the internal temperature stability of the spacecraft when it loses its heat source, consistent with the insulating function of natural honeycomb structures.

Therefore, based on the above analysis, the phase change latent heat of PCM effectively reduces the surface temperature when heat is loaded, improves the heat distribution, promotes heat conduction, and reduces the burden of convective heat transfer between the fuel coolant and channel wall in the convective cooling channel. At the same time, the phase change latent heat of PCM also plays a role in thermal insulation for the spacecraft as the system temperature drops. Therefore, the combination of PCM is an effective approach for designing convective cooling channels with excellent heat conduction capabilities.

5.2. Effect of Geometric Parameter on Heat Transfer Performance

5.2.1. Effect of Trapezoidal PCM Cross Section Characteristic Vector (W)

In this section, the effect of changing the characteristic vector of the trapezoidal PCM ($\mathbf{W_1} = (\frac{10}{3}, 5, \frac{6}{5}), \mathbf{W_2} = (\frac{5}{3}, \frac{20}{3}, \frac{6}{5}), \mathbf{W_3} = (4, 8, 1), \mathbf{W_4} = (2, 10, 1)$) in PCM-HC was assessed while keeping the PCM area and position unchanged. Other control parameters were kept constant, $Q_{in} = 50 \text{ kW/m}^2$.

The graph in Figure 11 illustrates the temperature variation along the EF path (upper wall) at the end of the thermal load (600 s) for different trapezoidal PCM cross section characteristic vectors under a heat flux input of 50 kW/m². It can be observed that the temperature distribution is symmetrical, and changes in characteristic vectors have minimal

impact on the temperature at the edges of the wall but affect the thermal distribution at the middle position of the wall. As the height h of the trapezoid increases, the temperature at the middle position decreases. Additionally, increasing the difference in length between the upper and lower bases of the trapezoid also leads to a decrease in the middle position temperature. When W = (2,10,1), the temperature difference between the highest and lowest temperatures at the middle position is 0.3 K. Although the overall dimensions of the model influence the magnitude of temperature differences, it also indicates that the trapezoidal PCM characteristic vector provides optimal heat dissipation. We also studied the influence of trapezoidal PCM cross section characteristic vectors on the thermal distribution and heat dissipation on the side and bottom walls. However, since the temperature hardly changes, we will not provide detailed explanations here.



Figure 11. The effect of trapezoidal PCM cross section characteristic vector change ($W_1 = (\frac{10}{3}, 5, \frac{6}{5})$, $W_2 = (\frac{5}{3}, \frac{20}{3}, \frac{6}{5})$, $W_3 = (4, 8, 1)$, $W_4 = (2, 10, 1)$) on the temperature distribution on the upper wall of PCM-HC under a heat flux input of 50 kW/m².

5.2.2. Effect of Belly Plate Thickness (N)

In this section, the effect of belly plate thickness (N = 0.15 mm, 0.30 mm, 0.45 mm, 0.60 mm) on heat transfer in PCM-HC will be investigated under the fixed trapezoidal PCM cross section characteristic vector of **W** = (2,10,1), while keeping the other control parameters constant, $Q_{\rm in} = 50$ kW/m².

Figure 12a illustrates that, under the same heat flux input, the temperature on the upper wall decreases with an increase in the baffle thickness. However, this decrease is not uniform; rather, it diminishes less as the thickness increases. This phenomenon arises because the increase in baffle thickness leads to an increase in the volume of the base SUS304 and a decrease in the PCM volume. SUS304 exhibits better thermal conductivity, facilitating faster heat transfer from the upper wall downwards. Simultaneously, the reduction in PCM volume reduces the heat absorption during PCM phase change, weakening its heat dissipation effect and thereby allowing heat to reach the side and lower walls more rapidly, resulting in a decrease in temperature on the upper wall. However, there exists a balance point, due to the limited heat conduction speed in the upper wall region, where the increase in baffle thickness causing accelerated heat conduction reaches equilibrium, hence reducing the decrease in temperature on the upper wall. The maximum temperature difference between baffle thicknesses of 0.60 mm and 0.15 mm is approximately 1.5 K.



Figure 12. The effect of belly plate thickness changes (N = 0.15, 0.30, 0.45, and 0.60 mm) on the heat transfer performance of PCM-HC ($Q_{in} = 50 \text{ kW/m}^2$, $\mathbf{W} = (2,10,1)$). (a) The temperature distribution variation on the upper wall (EF red line). (b) The temperature distribution variation on the side wall (GH red line). (c) The variation of the maximum temperature on the lower wall with increasing belly plate thickness *N*. (d) The temperature variation on the lower wall during the entire loading and unloading process after 600 s.

Additionally, in Figure 12b, under the same heat flux input, except for N = 0.60 mm, the temperature on the side wall decreases in the upper half section and increases in the lower half section with an increase in the baffle thickness. The maximum temperature difference between baffle thicknesses of 0.60 mm and 0.15 mm exceeds 0.5 K. The variation inside wall temperature distribution is analogous to that of the upper wall, wherein the combined effects of SUS304 and PCM result in non-uniform changes in heat distribution with changing baffle thickness. Increasing baffle thickness accelerates heat transfer towards the channel's far end, yet weakens PCM's attraction to heat flux on the upper wall and its hindrance to heat transfer on the lower wall simultaneously.

Figure 12c illustrates the trend of the maximum temperature on the lower wall of the PCM-HC channel as the baffle thickness increases, showing a nearly linear relationship. The temperature difference between N = 0.60 mm and N = 0.15 mm is 6 K. This is attributed to the constant thickness of the left and right support pillars on the side wall, resulting in an increase in PCM and a decrease in SUS304 as the baffle thickness decreases. Compared to SUS304, a larger area of PCM enhances the heat dissipation capacity of the channel. PCM possesses latent heat of phase change and a larger specific heat capacity, meaning

that when the area of the PCM-HC channel remains constant, PCM can absorb more heat at much lower temperatures. Simultaneously, PCM's latent heat of phase change absorbs more heat, resulting in a lower temperature on the lower wall.

Figure 12d depicts the variation trend of the lower wall temperature throughout the entire process of removing the heat load after 600 s of thermal loading under different baffle thicknesses in the PCM-HC channel. It can be observed that as the baffle thickness decreases, the heat dissipation effect of PCM, both with and without heat flux input, improves. Comparing N = 0.15 mm to N = 0.60 mm, with heat flux input, the maximum temperature on the lower wall decreases from 390.4 K to 389.5 K, and without heat flux input, it decreases from 415.1 K to 409.0 K. This indicates that adjusting the baffle thickness to enhance heat conduction on the wall of the convective cooling channel yields favorable results.

In summary, reducing the thickness of the baffle significantly decreases the temperature of the lower wall while causing a slight increase in temperature in certain regions of the upper and side walls. However, this increase is minimal compared to the temperature drop on the lower wall and can be considered negligible. Therefore, overall, reducing the baffle thickness is conducive to enhancing the heat dissipation capability of the PCM-HC channel. However, it is imperative to consider the mechanical strength of PCM-HC; decreasing the baffle thickness will inevitably diminish the channel's load-bearing performance. Consequently, the actual baffle thickness should be selected based on the specific load-bearing requirements of the channel. Hence, taking into account both heat transfer and load-bearing performance, a compromise is made in the test design, with a relatively optimal baffle thickness of 0.30 mm chosen.

5.3. Effect of PCM Species on Heat Transfer Performance

In this section, the effect of using different PCMs on PCM-HC heat transfer will be studied under the optimal geometric combination with a fixed trapezoidal PCM cross section characteristic vector of $\mathbf{W} = (2,10,1)$ and a fixed belly plate thickness of N = 0.30 mm, while keeping other control parameters constant and $Q_{in} = 50$ kW/m². To achieve better thermal management performance, PCMs with low thermal conductivity, high latent heat of phase change, and high specific heat capacity are preferred, provided that the PCM phase change temperature is lower than the actual temperature inside the system. Erythritol, Urea, and Paraffin CPCM were selected as the three PCM options, and their thermophysical properties [64–66] are listed in Table 2.

Table 2. PCM thermophysical properties.

Property	Paraffin CPCM	Erythritol	Urea
Density (kg/m ³)	780	1450	1320
Specific heat capacity (J/kg/K)	2350	2610	2100
Thermal conductivity (W/m/K)	1.12	0.73	0.8
Phase transition temperature (K)	343.15	391.15	407.15
Latent heat (kJ/kg)	189.6	340	250

As shown in Figure 13a,b, the filling of different PCMs has produced varying effects on the temperature distribution of the upper and side walls of PCM-HC. Erythritol, with its higher specific heat capacity and phase change heat and lower thermal conductivity compared to the other two PCMs, exhibits excellent thermal conductivity. Compared to the HC without PCM, the average temperature of the upper wall decreases by 12 K, and the average temperature of the side wall decreases by 15 K when using Erythritol. On the other hand, Urea shows poor thermal conductivity due to its higher phase change temperature relative to Erythritol and lower phase change latent heat. Within the limited heat flux loading time, the upper wall completes the phase change latent heat process early and continues to release sensible heat, while some regions of the PCM in the side wall do not reach the phase change temperature, resulting in the incomplete utilization of PCM's thermal conductivity ability. As a result, the average temperature of the upper wall increases by 1.2 K, and the temperature of the side wall decreases by 0.5 K. The addition of Paraffin CPCM has the opposite effect on the system, leading to an increase in the temperatures of both the upper and side walls. The average temperature of the upper wall rises by nearly 11 K, and the average temperature of the side wall increases by 7.5 K. This is because Paraffin CPCM has a lower phase change temperature. At $Q_{in} = 50 \text{ kW/m}^2$ and within a loading time of 600 s, PCM-HC completes the phase change and releases sensible heat much earlier. Additionally, the highest temperature on the lower wall of the PCM-HC channel in Figure 13c reflects the different effects of the three PCMs analyzed above. Compared to the HC without PCM, Erythritol, Urea, and Paraffin CPCM exhibit temperature changes of a decrease by 21.8 K, a decrease by 7.6 K, and an increase by 9 K, respectively.



Figure 13. The effect of different PCMs (Erythritol, Urea, and Paraffin CPCM) on the heat transfer properties of PCM-HC ($Q_{in} = 50 \text{ kW/m}^2$, $\mathbf{W} = (2,10,1)$, N = 0.30 mm). (a) The variation of temperature distribution on the upper wall (EF red line). (b) The variation of temperature distribution on the side wall (GH red line). (c) Comparison of the highest temperature on the bottom surface for different PCMs. (d) Variation trend of the temperature on the lower wall during the entire process of removing the heat load after 600 s of loading with different PCM fillings.

Figure 13d shows the variation trend of the temperature on the lower wall of PCM-HC during the entire process of removing the heat load after 600 s of loading with different PCM fillings. A comparison reveals that Erythritol exhibits the best thermal conductivity of the PCM-HC structure. During the heat flux loading stage, the highest temperature on the bottom surface decreases by 13 K. In the absence of heat flux loading, the highest temperature on the bottom surface decreases by 20 K. Meanwhile, during the phase change process without heat flux input, the temperature remains almost unchanged for a certain period of time, which is approximately 2000 s for Erythritol, 1200 s for Urea, and 950 s for Paraffin CPCM. This indicates a delayed temperature decrease and demonstrates that Erythritol provides the best insulation effect among the three PCMs.

Therefore, the analysis of heat transfer performance for different PCMs demonstrates that not all PCMs are beneficial for improving the thermal conductivity of PCM-HC. In summary, when using PCM-HC, it is necessary to select a PCM with suitable phase change temperature and latent heat according to the structural parameters and material properties of TPS, as well as the size and duration of the input heat flux load.

6. Conclusions

In this paper, a novel convective cooling channel filled with PCM is proposed. The introduction of PCM aims to improve the heat dissipation capability of the channel, that is, to reduce the wall temperature of the channel. Firstly, a model of a honeycomb-like convective cooling channel (HC) was proposed. Then, two-dimensional topology optimization of HC was conducted using the variable density method based on the SIMP interpolation model, and the PCM-filled PCM-HC structure was constructed. The mechanical load-bearing capacities of HC, PCM-HC, and square channels are then compared. Finally, numerical analysis of heat transfer was carried out from three aspects, i.e., with and without PCM, variation of PCM geometric parameters, and use of different types of PCMs, to determine the optimal geometric combination of the designed convective cooling channel (trapezoidal PCM cross section characteristic vector of (2,10,1), and a thickness of 0.15 mm for the belly plate). The main conclusions are as follows: (1) The convective cooling channel with PCM filling exhibits excellent thermal conductivity. The melting phase change of PCM changes the heat distribution, effectively reducing the wall temperature of the channel and reducing the burden of convective heat transfer between fuel and channel wall. Under the maximum heat flux load input of the tested design, the highest temperature on the inner wall of the channel decreased by 12 K to 20 K. Meanwhile, during the process of PCM solidification latent heat, the temperature remains almost constant when the temperature of the spacecraft system drops, providing insulation for the spacecraft. (2) When the heat flux load remains constant and the area of the trapezoidal PCM cross section is kept unchanged, the temperature distribution on the upper wall of the PCM-HC channel shows variations with changes in the characteristic vector parameters. The optimal thermal conductivity of the channel's upper wall is achieved when the trapezoidal PCM cross section characteristic vector parameters are set to (2,10,1). However, due to the constraints of the overall size of the channel design, this parameter has a less significant impact on the heat transfer performance of the side wall and lower wall. (3) As the thickness of the belly plate decreases from N = 0.60 mm to N = 0.15 mm, the thermal conductivity of the convective cooling channel exhibits non-uniform variations. Reducing the thickness of the belly plate can significantly lower the temperature of the lower wall of the channel. However, due to the combined effect of the two materials, SUS304 and PCM, in the channel, there is an uncertain impact on the temperature of the side walls. Nevertheless, this impact results in minimal temperature changes. Considering the load requirements of PCM-HC, a belly plate thickness of 0.30 mm is relatively optimal in the tested design, with the highest temperature on the lower wall decreasing by approximately 1 K. (4) Different types of PCMs have different effects on PCM-HC. Erythritol and Urea possess both thermal conductivity and insulation effects. However, Paraffin CPCM did not lower the wall temperature; instead, it increased it, attributed to the thermal physical properties of the PCM. (5) For the convective cooling

channel filled with PCM designed in this study for application in the TPS of hypersonic vehicles, it is advantageous for heat dissipation if the difference between the heights of trapezoidal PCM cross-section vectors and the lengths of their upper and lower bases is designed to be greater. Additionally, a smaller thickness of the belly plate is favorable for heat dissipation, provided it meets the channel's load-bearing requirements. Furthermore, the selection of PCM with appropriate phase change temperature and latent heat should be based on the specific operating conditions of the channel, such as the magnitude of the input heat flux load and the duration of loading, to ensure the full utilization of PCM's heat dissipation capabilities.

Author Contributions: Conceptualization, W.L.; methodology, W.L.; investigation, W.L. and X.W.; data curation, W.L., L.L. and X.W.; writing—original draft preparation, W.L.; writing—review and editing, J.Z., X.W., L.L., W.W. and S.Y.; supervision, J.Z.; funding acquisition, J.Z. All authors have read and agreed to the published version of the manuscript.

Funding: This work was supported by the National Key R&D Program of China (2021YFB3400200), the National Natural Science Foundation of China (grant 52075038, 52375282), the National Science and Technology Major Project of China (grant SKS-2022031), and the BIT Teli Young Fellow Recruitment Program (RCPT-20220005).

Data Availability Statement: The research data for this article can be obtained from the corresponding author "Jieliang Zhao" on reasonable request.

Conflicts of Interest: The authors declare no conflict of interest.

References

- 1. Schmisseur, J.D. Hypersonics into the 21st Century: A Perspective on Afosr-Sponsored Research in Aerothermodynamics. *Prog. Aerosp. Sci.* 2015, 72, 3–16. [CrossRef]
- Sziroczak, D.; Smith, H. A Review of Design Issues Specific to Hypersonic Flight Vehicles. Prog. Aerosp. Sci. 2016, 84, 1–28. [CrossRef]
- Glass, D.E. Ceramic Matrix Composite (Cmc) Thermal Protection Systems (Tps) and Hot Structures for Hypersonic Vehicles. In Proceedings of the 15th AIAA International Space Planes and Hypersonic Systems and Technologies Conference, Dayton, OH, USA, 28 April–1 May 2008.
- 4. Liu, W.; Shu, B.; Guo, L.; He, H. Impacts of Thermal Environment on Modal Frequency of Fgm Shells. J. Vib. Shock. 2017, 36, 127–131.
- Jackson, T.A.; Eklund, D.R.; Fink, A.J. High Speed Propulsion: Performance Advantage of Advanced Materials. *J. Mater. Sci.* 2004, 19, 39. [CrossRef]
- 6. Yang, Q.; Xie, W.; Peng, Z.; Meng, S.; Du, S. New Concepts and Trends in Development of Thermal Protection Design and Analysis Technology. *Hangkong Xuebao/Acta Aeronaut. Astronaut. Sinica* **2015**, *36*, 2981–2991.
- Allouhi, A.; Agrouaz, Y.; Amine, M.B.; Rehman, S.; Buker, M.S.; Kousksou, T.; Jamil, A.; Benbassou, A. Design Optimization of a Multi-Temperature Solar Thermal Heating System for an Industrial Process. *Appl. Energy* 2017, 206, 382–392. [CrossRef]
- 8. Barth, T.; Longo, J.M.A. Advanced Aerothermodynamic Analysis of Shefex I. Aerosp. Sci. Technol. 2010, 14, 587–593. [CrossRef]
- 9. Chang, B.; Huang, J.; Yao, W.-X. Thermal Protection Mechanism of a Novel Adjustable Non-Ablative Thermal Protection System for Hypersonic Vehicles. *Aerospace* 2023, 10, 1. [CrossRef]
- Rivers, H.K.; Glass, D.E. Advances in Hot-Structure Development. In Proceedings of the 5th European Workshop on Thermal Protection Systems and Hot Structures, Noordwiji, The Netherlands, 17–19 May 2006.
- 11. Bao, W.; Li, X.L.; Qin, J.; Zhou, W.X. Modelling and Simulation Methodology of Channel Cooling Using Hydrocarbon Fuel as Coolant under Supercritical Pressures. *Proc. Inst. Mech. Eng. Zhou Part G J. Aerosp. Eng.* **2011**, 225, 969–984. [CrossRef]
- 12. Zhang, S.; Qin, J.; Xie, K.; Feng, Y.; Bao, W. Thermal Behavior inside Scramjet Cooling Channels at Different Channel Aspect Ratios. *J. Propuls. Bao Power* **2016**, *32*, 57–70. [CrossRef]
- 13. Uyanna, O.; Najafi, H. Thermal Protection Systems for Space Vehicles: A Review on Technology Development, Current Challenges and Future Prospects. *Acta Astronaut.* 2020, *176*, 341–356. [CrossRef]
- 14. Maruyama, S.; Viskanta, R.; Aihara, T. Active Thermal Protection System against Intense Irradiation. *J. Thermophys. Heat Transf.* **2012**, *3*, 389–394. [CrossRef]
- Liu, S.; Zhang, B. Effects of Active Cooling on the Metal Thermal Protection Systems. *Aerosp. Sci. Technol.* 2011, 15, 526–533. [CrossRef]
- Zhou, W.; Johnson, B.; Hu, H. Effects of Flow Compressibility and Density Ratio on Film Cooling Performance. J. Propuls. Power 2016, 33, 964–974. [CrossRef]
- 17. Lee, Y.J.; Kim, S.J. Thermal Optimization of the Pin-Fin Heat Sink with Variable Fin Density Cooled by Natural Convection. *Appl. Therm. Eng.* **2021**, 190, 116692. [CrossRef]

- Vinh Tung, L.; San Ha, N.; Seo Goo, N. Advanced Sandwich Structures for Thermal Protection Systems in Hypersonic Vehicles: A Review. *Compos. Part B Eng.* 2021, 226, 109301.
- 19. Naraghi, M.; Dunn, S.; Coats, D. A Model for Design and Analysis of Regeneratively Cooled Rocket Engines. In Proceedings of the 40th AIAA/ASME/SAE/ASEE Joint Propulsion Conference and Exhibit, Fort Lauderdale, FL, USA, 11–14 July 2004.
- 20. Paquette, E.T.; Warburton, R.; Eckel, A.; Jaskowiak, M. Cooled Cmc Scramjet Combustor Structure Development. In Proceedings of the 38th AIAA/ASME/SAE/ASEE Joint Propulsion Conference & Exhibit, Indianapolis, IN, USA, 7–10 July 2002.
- 21. Paquette, E. Cooled Cmc Structures for Scramjet Engine Flowpath Components. In Proceedings of the Aiaa/cira International Space Planes & Hypersonics Systems & Technologies Conference, Capua, Italy, 16–20 May 2006.
- 22. Eckert, E.R.G.; Livingood, J.N.J. Comparison of Effectiveness of Convection-, Transpiration-, and Film-Cooling Methods with Air as Coolant; National Advisory Committee for Aeronautics: Washington, DC, USA, 1953; Volume 1182.
- 23. Sadique, H.; Murtaza, Q. Samsher Heat Transfer Augmentation in Microchannel Heat Sink Using Secondary Flows: A Review. *Int. J. Heat Mass Transf.* **2022**, *194 Pt 2*, 123063. [CrossRef]
- 24. Wang, L.-H.; Liu, X.; Wang, H.-Y.; Liu, L.-Z. Study of the Technique of Phase Change Thermal Storage Applied in Aircraft Thermal Control System. *Missiles Space Veh.* **2017**, *6*, 88–92.
- 25. Olds, J.R. Results of a Rocket-Based Combined-Cycle Ssto Design Using Parametric Mdo Methods. J. Aerosp. 1994, 103, 154–173.
- 26. Winston, M.L. The Biology of the Honey Bee; Harvard University Press: Cambridge, MA, USA, 1991.
- 27. Karihaloo, B.L.; Zhang, K.; Wang, J. Honeybee Combs: How the Circular Cells Transform into Rounded Hexagons. J. R. Soc. Interface 2013, 10, 20130299. [CrossRef] [PubMed]
- Zhang, K.; Duan, H.; Karihaloo, B.L.; Wang, J. Hierarchical, Multilayered Cell Walls Reinforced by Recycled Silk Cocoons Enhance the Structural Integrity of Honeybee Combs. *Proc. Natl. Acad. Sci. USA* 2010, 107, 9502–9506. [CrossRef] [PubMed]
- 29. Zhang, K.; Si, F.W.; Duan, H.L.; Wang, J. Microstructures and Mechanical Properties of Silks of Silkworm and Honeybee. *Acta Biomater.* **2010**, *6*, 2165–2171. [CrossRef] [PubMed]
- Duan, H.L.; Wang, J.; Karihaloo, B.L.; Huang, Z.P. Nanoporous Materials Can Be Made Stiffer Than Non-Porous Counterparts by Surface Modification. Acta Mater. 2006, 54, 2983–2990. [CrossRef]
- Duan, H.L.; Karihaloo, B.L. Thermo-Elastic Properties of Heterogeneous Materials with Imperfect Interfaces: Generalized Levin's Formula and Hill's Connections. J. Mech. Phys. Solids 2007, 55, 1036–1052. [CrossRef]
- 32. Lu, T.J. Heat Transfer Efficiency of Metal Honeycombs. Int. J. Heat Mass Transf. 1999, 42, 2031–2040. [CrossRef]
- 33. Wen, T.; Tian, J.; Lu, T.J.; Queheillalt, D.T.; Wadley, H.N.G. Forced Convection in Metallic Honeycomb Structures. *Int. J. Heat Mass Transf.* 2006, *49*, 3313–3324. [CrossRef]
- Pei, B.; Chen, Z.; Li, F.; Bai, B. Flow and Heat Transfer of Supercritical Co2 in the Honeycomb Ultra-Compact Plate Heat Exchanger. J. Supercrit. Fluids 2019, 148, 1–8. [CrossRef]
- 35. Nkwetta, D.N.; Haghighat, F. Thermal Energy Storage with Phase Change Material—A State-of-the Art Review. *Sustain. Cities Soc.* 2014, *10*, 87–100. [CrossRef]
- 36. Kim, T.Y.; Hyun, B.-S.; Lee, J.-J.; Rhee, J. Numerical Study of the Spacecraft Thermal Control Hardware Combining Solid–Liquid Phase Change Material and a Heat Pipe. *Aerosp. Sci. Technol.* **2013**, *27*, 10–16. [CrossRef]
- Wang, S.; Zhang, D.; Li, C.; Wang, J.; Zhang, J.; Cheng, Y.; Mei, W.; Cheng, S.; Qin, P.; Duan, Q. Numerical Optimization for a Phase Change Material Based Lithium-Ion Battery Thermal Management System. *Appl. Therm. Eng.* 2023, 222, 119839. [CrossRef]
- Tang, L.; Gao, T.; Song, L.; Meng, L.; Zhang, C.; Zhang, W. Topology Optimization of Nonlinear Heat Conduction Problems Involving Large Temperature Gradient. *Comput. Methods Appl. Mech. Eng.* 2019, 357, 112600. [CrossRef]
- 39. Yu, M.; Ruan, S.; Gu, J.; Ren, M.; Li, Z.; Wang, X.; Shen, C. Three-Dimensional Topology Optimization of Thermal-Fluid-Structural Problems for Cooling System Design. *Struct. Multidiscip. Optim.* **2020**, *62*, 3347–3366. [CrossRef]
- 40. Fawaz, A.; Hua, Y.; Le Corre, S.; Fan, Y.; Luo, L. Topology Optimization of Heat Exchangers: A Review. *Energy* **2022**, 252, 124053. [CrossRef]
- Sigmund, O.; Maute, K. Topology Optimization Approaches: A Comparative Review. Struct. Multidiscip. Optim. 2013, 48, 1031–1055. [CrossRef]
- 42. Zhuang, C.; Xiong, Z.; Ding, H. Topology Optimization of the Transient Heat Conduction Problem on a Triangular Mesh. *Numer. Heat Transf. Part B Fundam.* 2013, 64, 239–262. [CrossRef]
- 43. Xia, Q.; Shi, T.; Xia, L. Topology Optimization for Heat Conduction by Combining Level Set Method and BESO Method. *Int. J. Heat Mass Transf.* **2018**, 127, 200–209. [CrossRef]
- 44. Zhou, M.; Geng, D. Multi-Scale and Multi-Material Topology Optimization of Channel-Cooling Cellular Structures for Thermomechanical Behaviors. *Comput. Methods Appl. Mech. Eng.* 2021, 383, 113896. [CrossRef]
- 45. Li, Q.; Xu, R.; Liu, S.; Liang, G.; Qu, Y. Topology Optimization Design of Multi-Material Quasi-Periodic Cellular Structures for Thermoelastic Responses. *Int. J. Numer. Methods Eng.* **2022**, 123, 4345–4366. [CrossRef]
- Huang, M.; Sha, W.; Xiao, M.; Gao, L.; Liu, X.; Wang, Y. Multiscale Topology Optimization of Cellular Structures with High Thermal Conductivity and Large Convective Surface Area. *Int. J. Therm. Sci.* 2024, 201, 109053. [CrossRef]
- 47. Tautz, J.; Heilmann, H.R. Phänomen Honigbiene; Springer: Berlin/Heidelberg, Germany, 2007.
- 48. Sigmund, O. A 99 Line Topology Optimization Code Written in Matlab. Struct. Multidiscip. Optim. 2001, 21, 120–127. [CrossRef]
- 49. Rietz, A. Sufficiency of a Finite Exponent in Simp (Power Law) Methods. Struct. Multidiscip. Optim. 2001, 21, 159–163. [CrossRef]

- 50. Myers, D.E. Parametric Weight Comparison of Advanced Metallic, Ceramic Tile, and Ceramic Blanket Thermal Protection Systems; National Aeronautics and Space Administration, Langley Research Center: Hampton, VA, USA, 2000.
- 51. Ma, Y.; Xu, B.; Chen, M.; He, R.; Wen, W.; Cheng, T.; Fang, D. Optimization Design of Built-Up Thermal Protection System Based on Validation of Corrugated Core Homogenization. *Appl. Therm. Eng.* **2017**, *115*, 491–500. [CrossRef]
- 52. Yang, X.-H.; Tan, S.-C.; He, Z.-Z.; Zhou, Y.-X.; Liu, J. Evaluation and Optimization of Low Melting Point Metal Pcm Heat Sink against Ultra-High Thermal Shock. *Appl. Therm. Eng.* **2017**, *119*, 34–41. [CrossRef]
- 53. Sánchez, P.S.; Ezquerro, J.M.; Fernandez, J.; Rodriguez, J. Thermocapillary Effects During the Melting of Phase Change Materials in Microgravity: Heat Transport Enhancement. *Int. J. Heat Rodriguez Mass Transf.* **2020**, *163*, 120478. [CrossRef]
- 54. Chen, Y.; Zhuang, Y.; Feng, J.-C.; Huang, S.-M. Numerical Investigation on Flow and Heat Transfer Characteristics in Honeycomb-Like Microchannel Heat Sink Encapsulated with Phase Change Material. *Appl. Therm. Eng.* **2023**, 232, 121060. [CrossRef]
- 55. Bhuiya, R.; Shah, N.; Arora, D.; Krishna, N.V.; Manikandan, S.; Selvam, C.; Lamba, R. Thermal Management of Phase Change Material Integrated Thermoelectric Cooler with Different Heat Sink Geometries. *J. Energy Storage* **2022**, *51*, 104304. [CrossRef]
- 56. Ren, X.M.; Nie, H.; Zhang, M.; Nikitchenko, Y.A.; Tikhonovets, A. Heat Performance Investigation of Phase Change Materials for the Cooling Array System on Vehicles. *Appl. Therm. Eng.* **2020**, *171*, 115115. [CrossRef]
- 57. Voller, V.R. An Overview of Numerical Methods for Solving Phase Change Problems. Adv. Numer. Heat Transf. 1997, 1, 341–380.
- 58. Niezgoda-Żelasko, B. The Enthalpy-porosity Method Applied to the Modelling of the Ice Slurry Melting Process During Tube Flow. *Procedia Eng.* **2016**, 157, 114–121. [CrossRef]
- 59. Hasis, F.B.A.; Krishna, P.M.; Aravind, G.P.; Deepu, M.; Shine, S.R. Thermo Hydraulic Performance Analysis of Twisted Sinusoidal Wavy Microchannels. *Int. J. Therm. Sci.* 2018, 128, 124–136. [CrossRef]
- 60. Debich, B.; El Hami, A.; Yaich, A.; Gafsi, W.; Walha, L.; Haddar, M. Design Optimization of Pcm-Based Finned Heat Sinks for Mechatronic Components: A Numerical Investigation and Parametric Study. J. Energy Storage 2020, 32, 101960. [CrossRef]
- 61. Yang, Y.-T.; Wang, Y.-H. Numerical Simulation of Three-Dimensional Transient Cooling Application on a Portable Electronic Device Using Phase Change Material. *Int. J. Therm. Sci.* **2012**, *51*, 155–162. [CrossRef]
- Egolf, P.W.; Manz, H. Theory and Modeling of Phase Change Materials with and without Mushy Regions. Int. J. Heat Mass Transf. 1994, 37, 2917–2924. [CrossRef]
- 63. Wang, Y.-H.; Yang, Y.-T. Three-Dimensional Transient Cooling Simulations of a Portable Electronic Device Using PCM (Phase Change Materials) in Multi-Fin Heat Sink. *Energy* **2011**, *36*, 5214–5224. [CrossRef]
- 64. Sarbu, I.; Dorca, A. Review on Heat Transfer Analysis in Thermal Energy Storage Using Latent Heat Storage Systems and Phase Change Materials. *Int. J. Energy Res.* 2018, 43, 29–64. [CrossRef]
- Zalba, B.; Marín, J.M.; Cabeza, L.F.; Mehling, H. Review on Thermal Energy Storage with Phase Change: Materials, Heat Transfer Analysis and Applications. *Appl. Therm. Eng.* 2003, 23, 251–283. [CrossRef]
- Solé, A.; Neumann, H.; Niedermaier, S.; Martorell, I.; Schossig, P.; Cabeza, L.F. Stability of Sugar Alcohols as PCM for Thermal Energy Storage. Sol. Energy Mater. Sol. Cells 2014, 126, 125–134. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.